Author’s response to reviews

Title: Inhibition of advanced glycation end products by red grape skin extract and its antioxidant activity

Authors:

Nattha Jariyapamornkoon (Fhay_vet@yahoo.com)
Sirintorn Yibchok-anun (Sirintorn.y@chula.ac.th)
Sirichai Adisakwattana (Sirichai.a@chula.ac.th)

Version: 2 Date: 20 February 2013

Author’s response to reviews: see over
Dear Sir:

I, along with my coauthors, would like to ask you to consider the attached manuscript entitled “Inhibition of advanced glycation end products by red grape skin extract and its antioxidant activity” for publication in BMC complementary and alternative medicine as a research article.

We believe that the findings of this study are relevant to the scope of your journal and will be of interest to its readership.

This manuscript has not been published or presented elsewhere in part or in entirety, and is not under consideration by another journal. All the authors have approved the manuscript and agree with submission to your esteemed journal. There are no conflicts of interest to declare.

I look forward to hearing from you at your earliest convenience.

Sincerely,
Sirichai Adisakwattana
Department of Nutrition and Dietetics
Faculty of Allied Health Sciences
Chulalongkorn University
Tel.: +662-218-1067
Fax: +662-218-1076
Email: Sirichai.a@chula.ac.th